Author’s response to reviews

Title: The association between RCAS1 expression in laryngeal and pharyngeal cancer and its healthy stroma with cancer relapse

Authors:

Magdalena Dutsch-Wicherek (mowicher@gmail.com)
Romana Tomaszewska (romatom@mp.pl)
Agata Lazar (agalaz@interia.pl)
Jacek Skladzien (jaskladzien@cm-uj.krakow.pl)
Lukasz Wicherek (mowicher@cyf-kr.edu.pl)

Version: 7 Date: 25 January 2009

Author’s response to reviews: see over
Jan 24, 2008

BMC Cancer

Editor-in-Chief

Dear Editor-in-Chief

I would like to resubmit a corrected version of the manuscript entitled: “The association between RCAS1 expression in laryngeal and pharyngeal cancer and its healthy stroma with cancer relapse” to the BMC Cancer.

With best regards,

Magdalena Dutsch-Wicherek, M.D., Ph.D

Otolaryngology Department

Jagiellonian University

Krakow, Poland

mowicher@cyf-kr.edu.pl
Below, please find the list of changes:

1. Page 6, line 4, the sentence: “Heparinized blood samples were centrifuged first at 2000g, and then at 16000g. The supernatants containing blood serum were mixed with the proteinase inhibitor cocktail and SDS buffer as described above” was removed.

2. Page 10, line 22, the word “Line” was corrected into line

3. Should the headings of the tables be highlighted with bold or not? Because in the manuscript that I have received the headings of the following tables: 2, 6 were highlighted with bold, when the other headings were not. I have not corrected it.

4. Table 2: I have formatted the table. The word: “immunorectivity” was corrected into “immunoreactivity” in the comment below the table.